Author's response to reviews

Title: Evidence-based interventions in dementia: A pragmatic cluster-randomised trial of an educational intervention to promote earlier recognition and response to dementia in primary care (EVIDEM-ED)

Authors:

Steve Iliffe Prof (s.iliffe@pcps.ucl.ac.uk)
Jane Wilcock Ms (j.wilcock@pcps.ucl.ac.uk)
Mark Griffin Mr (m.griffin@pcps.ucl.ac.uk)
Priya Jain Mrs (p.jain@pcps.ucl.ac.uk)
Ingela Thuné-Boyle Dr (i.thune-boyle@pcps.ucl.ac.uk)
Tamar Koch Dr (t.koch@ucl.ac.uk)
Frances Lefford Dr (f.lefford@pcps.ucl.ac.uk)

Version: 2 Date: 19 January 2010

Author's response to reviews:

Thank you for the helpful feedback. We have added details of the ethics committee approval, and included a justification of the effect size used in the sample size calculation.

Steve Iliffe, for the authors